Author's response to reviews

Title: A new tool to evaluate postgraduate training posts: the Job Evaluation Survey Tool (JEST)

Authors:

David Wall (d.w.wall.fife@btinternet.com)
Helen M Goodyear (Helen.Goodyear@wm.hee.nhs.uk)
Baldev Singh (Baldev.Singh@nhs.net)
Andrew Whitehouse (Andrew.Whitehouse@wm.hee.nhs.uk)
Elizabeth Hughes (Elizabeth.Hughes@wm.hee.nhs.uk)
Jonathan Howes (Jonathan.Howes@wm.hee.nhs.uk)

Version: 2 Date: 25 January 2014

Author's response to reviews:

WE have added a section entitled Acknowledgements and funding. It is important to note that there are no acknowledgements and the paper was written by the authors with no other scientific writer involved. There was no funding for this study which was undertaken in the authors unfunded time.